Author's response to reviews

Title: Anti-inflammatory and antioxidant effects of Tualang honey in alkali injury on the eyes of rabbits: Experimental animal study

Authors:

Karuppannan Bashkaran (bashkaran@yahoo.com)
Embong Zunaina (zunaina@kb.usm.my)
Shaharuddin Bakiah (bakiah@kb.usm.my)
Siti-Amrah Sulaiman (sbamrah@kb.usm.my)
KNS Sirajudeen (sirajuden@kb.usm.my)
Venkatesh Naik (venkatesh@kb.usm.my)

Version: 2 Date: 19 April 2011

Author's response to reviews: see over
Dear Editor,

MS: 6827016695395258
Research article
Anti-inflammatory and antioxidant effects of Tualang honey in alkali injury on the eyes of rabbits: Experimental animal study
Karuppannan Bashkaran, Embong Zunaina, Shaharuddin Bakiah, Siti-Amrah Sulaiman, KNS Sirajudeen and Venkatesh Naik

Response to Editorial

Comment
Methods section – Please document ethical approval for the animal experiments and include the details of the institutional ethics committee that granted approval for the study.

Response
We have added the ethical approval in the methods section to the manuscript.

(page 6, paragraph 5)

**Ethical approval**
This study was approved by the Research and Ethical Committee, Health Campus, University Sains Malaysia on 16th March 2009 and also by Animal Ethics Committee, Health Campus, Universiti Sains Malaysia on 7th May 2009.

Yours sincerely,

Dr Zunaina Embong
Ophthalmologist and Medical lecturer,
Department of Ophthalmology,
School of Medical Sciences
Health Campus
Universiti Sains Malaysia
16150 Kota Bharu
Kelantan
Malaysia